\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{www.ijrps.com}{\color{blue}\underline{\smash{www.ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Shrikrishna D Gite et al.}
\firstPage{2265}
\articleType{Original Article}
\receivedDate{06 Dec 2019}
\acceptedDate{08 Feb 2020}
\revisedDate{30 Jan 2020}
\journalVolume{2020, 11}
\journalIssue{2}
\journalDoi{ijrps.v11i2.2187}
\copyrightYear{2020}
\def\authorCount{4}
\def\affCount{3}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Development and evaluation of colon targeted delivery of budesonide polymeric nanoparticles for colitis therapy}
\author{Shrikrishna~D~Gite\textsuperscript{*}\textsuperscript{1},  
            Kishor~S~Salunkhe\textsuperscript{2},  
            Ravindra~B~Chintamani\textsuperscript{1}, 
            Shubhangi~S~G\textsuperscript{3}~\\[5pt]{\textsuperscript{1}Research Scholar, Amrutvahini College of Pharmacy, Sangamner, Maharashtra, India}~\\{\textsuperscript{2}Department of Pharmaceutics, Sanjivani College of Pharmaceutical Education and Research, Kopargaon, Maharashtra, India}~\\{\textsuperscript{3}Department of Pharmaceutics, Vidyabharati College of Pharmacy, Amravati, Maharashtra, India}}

\begin{abstract}
Targeted delivery of the drug at site of action in case of Inflammatory Bowel Disease like colitis is the big challenge for formulators. The case where conventional drug delivery fails in severe stages of Inflammatory Bowel Disease, Nanoparticles is a good dosage form to targeted inflammatory site. The idea of this research was to prepare Nanoparticles with polymer Eudragit S100 and Surfactant Poloxamer containing Budesonide which finally filled in capsules providing immediate release at the ileocecal site, the most affected area in IBD i.e. pH dependent release. Nanoparticles are prepared by nano precipatation technique. Budesonide was used as a drug because of its therapeutic potential for in IBD. This study compares the different ratios of drug to olymer and drug to surfactant with optimized solvent and anti-solvent concentration in preparation of Nanoparticles. Optimized formulation ratio of drug to polymer was 1:2 and drug to surfactant was 1:1. The optimized batch of nanoparticles was filled in capsule and was evaluated for in-vitro studies.\textbf{\space }
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Budesonide,\newline Capsule,\newline Eudragit S100,\newline Colitis,\newline Nanoparticles
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Shrikrishna~D~Gite~\\ Phone:\ +91-8830691436~\\  Email:\ gitesd.pharma@gmail.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{www.ijrps.com}{www.ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
Major population worldwide is getting affected by Inflammatory Bowel Disease (IBD) and the prevalence of the disease is increasing day by day \unskip~\citep{687426:16266315}. This colitis (inflammation of colon) etiology and pathogenesis are not completely understood. Symptoms of IBD like ulceration, bloody diarrhoea, weight loss and which further lead to complete obstruction of the gastrointestinal (GI) tract, affect very badly on day today life of people \unskip~\citep{687426:16266121}. Most accepted move towards treating IBD and to reduce the need for surgeries and hospitalizations is by facilitating mucosal healing; induce remission of outbreaks \unskip~\citep{687426:16266325}. Systemic corticosteroids and mesalazine are used to treat Symptoms of mild-to-moderate IBD \unskip~\citep{687426:16265673}. These drugs mentioned above are preferred in many cases of pharmacotherapy for colon inflammation consisting of permanent administration of either of drug. It is necessary to take in consideration the quality and severity of adverse effects of these therapeutic regimens. For successful IBD treatment, it is necessary to involve innovative strategies for exact and adequate drug delivery to the inflammed colon for a long-lasting period in a sustained manner to minimize the risk of systemic adverse effects \unskip~\citep{687426:16266079}. For efficient treatment of colonic disorder, colon precise drug delivery is necessary. Nowadays, it is more complicated to target drug delivery to colon, so new techniques like nanotechnology are emerging out. Nanotechnology is used to aim description, manufacture and application of structure, having nanometer size scale. According to the direction of International System of Units (SI) nanotechnology is typically measured in nanometers scale of 1 billionth of a meter referred as the ``tiny science''. Minute size molecules and particle act in a different way, as an intact unit in requisites to provide a variety of advantages. Nanoparticles (NPs) are distinct particles of size 1 and 100 nanometres drug transporter.


\begin{table*}[!htbp]
\caption{\boldmath {Drug excipient ratio for compatibility study} }
\label{tw-39334c4f6b32}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4752\linewidth-2\tabcolsep}p{\dimexpr.1971999999999999\linewidth-2\tabcolsep}p{\dimexpr.3276\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Drug + Excipient & Ratio  & Total Weight (mg)\\
\tblmidrule 
Budesonide + Eudragit S100  &
  1:1  &
  20\\
Budesonide + Poloxamer  &
  1:1  &
  20\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Compatibility study of drug-Excipient mixture} }
\label{tw-3a4517d34f62}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3797\linewidth-2\tabcolsep}p{\dimexpr.6203\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Days & Compatibility results after 14 day study\\
\tblmidrule 
Drug + Eudragit S100 &
  Drug is compatible with Eudragit S100\\
Drug + Poloxamer &
  Drug is compatible with Poloxamer\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Design of Trial Batches} }
\label{tw-8d3a15514c51}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{LLLLLLL}
\tbltoprule \rowcolor{kwdboxcolor}Batch & Drug \mbox{}\protect\newline (mg) \mbox{}\protect\newline Budesonide & Polymer \mbox{}\protect\newline (mg) \mbox{}\protect\newline ES100 & Surfactant \mbox{}\protect\newline (mg) \mbox{}\protect\newline Poloxamer & Solvent \mbox{}\protect\newline (ml) \mbox{}\protect\newline Ethanol & Anti-solvent (ml) \mbox{}\protect\newline Water & Result\\
\tblmidrule 
F1 &
  50 &
  100 &
  50 &
  15 &
  30 &
  Particles formed\\
F2 &
  50 &
  75 &
  50 &
  15 &
  30 &
  Particles not formed\\
F3 &
  50 &
  50 &
  50 &
  15 &
  30 &
  Particles not formed \\
F4 &
  50 &
  100 &
  30 &
  15 &
  30 &
  Particles formed\\
F5 &
  50 &
  75 &
  30 &
  15 &
  30 &
  Particles not formed\\
F6 &
  50 &
  50 &
  30 &
  15 &
  30 &
  Particles not formed \\
F7 &
  50 &
  100 &
  50 &
  15 &
  15 &
  Particles formed\\
F8 &
  50 &
  75 &
  50 &
  15 &
  15 &
  Particles not formed\\
F9 &
  50 &
  50 &
  50 &
  15 &
  15 &
  Particles not formed \\
F10 &
  50 &
  100 &
  30 &
  15 &
  15 &
  Particles not formed\\
F11 &
  50 &
  75 &
  30 &
  15 &
  15 &
  Particles not formed\\
F12 &
  50 &
  50 &
  30 &
  15 &
  15 &
  Particles not formed \\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}
The drug is attached to a Nanoparticles matrix. Nanospheres and nanocapsules collectively called as nanoparticles. In nanocapsules drug is restricted to a cavity which is enclosed by a distinct polymer membrane, and in nanospheres consist of matrix systems in which the drug is physically and uniformly dispersed. Where conventional techniques reach their limits, nanotechnology provides opportunities for the medical applications \unskip~\citep{687426:16266316}. For colon specific drug delivery, now a days we can make use of Eudragit polymers (RS100, L100 and S100) or cellulose acetate phthalate, which are soluble at or above pH 7, or dissolving at pH 6 respectively. In the current study, we used Eudragit S100 to manufacture micro carriers for a definite colon release of Budesonide. Eudragit, RS100 and RL100 are copolymers based on acrylic and methacrylic acid esters, which has a low level of quaternary ammonium group. As compared to Eudragit -RL (8.8-12\% Ammonium groups) Eudragit -RS has a lesser content of charged groups (4.5-6.8\%), due to which it is less permeable to water. Due to insolubility at physiological pH the co-polymers will swell form permeable film which will facilitate to manufacture sustained release formulation \unskip~\citep{687426:16265756,687426:16266207} drag it is used in various studies for colon targeting since low content in ammonium group permitting a low solubility in gastric fluid.
    
\section{Materials and Methods}
\textbf{Materials}

Budesonide was obtained from Wockhardt Ltd. Aurangabad. Eudragit S100 was obtained from Evonik Pvt. Ltd. Poloxamer from Mody Chemi-Pharma. Ethanol and Methanol were obtained from Fisher Scientific Pvt. Ltd. Acetone from Fisher Scientific Pvt. Ltd. Hydrochloric Acid from Benzer multitech Pvt. Ltd. Tribasic Sodium Phosphate and Sodium Hydroxide from Merck Specialities Pvt. Ltd. All other raw materials used were of pharmaceutical grade.

\textbf{Method}

\textbf{Pre-formulation}

Preformulation parameters like Oraganoleptic properties, Solubility, Melting point for drug and excipients was determined. Compatibility between drug and excipient was determined using FT-IR spectroscopy. DSC was performed to check thermal properties of the drug.


\bgroup
\fixFloatSize{images/11f10a1a-6b4d-4f60-a6f8-538eec26c7aa-upicture1.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/11f10a1a-6b4d-4f60-a6f8-538eec26c7aa-upicture1.png}{\includegraphics{images/11f10a1a-6b4d-4f60-a6f8-538eec26c7aa-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Infra-Redspectrum of Budesonide}}
\label{f-ea9afac03b31}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/694d75d3-f72a-4d57-990c-5886fecd55f2-upicture2.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/694d75d3-f72a-4d57-990c-5886fecd55f2-upicture2.png}{\includegraphics{images/694d75d3-f72a-4d57-990c-5886fecd55f2-upicture2.png}}{}
\makeatother 
\caption{\boldmath {FT-IR of E S100}}
\label{f-417f9c099526}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/4d847240-50fc-431a-982e-afff5d49671c-upicture3.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/4d847240-50fc-431a-982e-afff5d49671c-upicture3.png}{\includegraphics{images/4d847240-50fc-431a-982e-afff5d49671c-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Infra-Red of Poloxamer}}
\label{f-cb21793ab7e6}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/b2965132-b389-4b87-9ac6-fa735a48a18f-upicture4.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/b2965132-b389-4b87-9ac6-fa735a48a18f-upicture4.png}{\includegraphics{images/b2965132-b389-4b87-9ac6-fa735a48a18f-upicture4.png}}{}
\makeatother 
\caption{\boldmath {DSC analysis of Budesonide}}
\label{f-8a1fb96ff16d}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/90c4d50f-ca93-48f0-aca3-35be19619719-upicture5.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/90c4d50f-ca93-48f0-aca3-35be19619719-upicture5.png}{\includegraphics{images/90c4d50f-ca93-48f0-aca3-35be19619719-upicture5.png}}{}
\makeatother 
\caption{\boldmath {FT-IR Drug+ES100}}
\label{f-341f93cda9ac}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/594a7d21-e9b6-4f8f-9626-9d63d8f5849a-upicture6.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/594a7d21-e9b6-4f8f-9626-9d63d8f5849a-upicture6.png}{\includegraphics{images/594a7d21-e9b6-4f8f-9626-9d63d8f5849a-upicture6.png}}{}
\makeatother 
\caption{\boldmath {FT-IR of Drug + Poloxamer}}
\label{f-5f4d211c11e8}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/defbf777-3b36-4577-a661-0e931c803bc9-upicture7.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/defbf777-3b36-4577-a661-0e931c803bc9-upicture7.png}{\includegraphics{images/defbf777-3b36-4577-a661-0e931c803bc9-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Calibration of Budesonide in Ethanol }}
\label{f-22f11909349d}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/323dd831-1263-4c88-8bf2-3aca05e3fe2a-upicture8.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/323dd831-1263-4c88-8bf2-3aca05e3fe2a-upicture8.png}{\includegraphics{images/323dd831-1263-4c88-8bf2-3aca05e3fe2a-upicture8.png}}{}
\makeatother 
\caption{\boldmath {Calibration ofBudesonide in 0.1N HCL}}
\label{f-c029e03bfb41}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/f2fdbded-57ba-4a93-9697-6d010859e52d-upicture9.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/f2fdbded-57ba-4a93-9697-6d010859e52d-upicture9.png}{\includegraphics{images/f2fdbded-57ba-4a93-9697-6d010859e52d-upicture9.png}}{}
\makeatother 
\caption{\boldmath {Calibration of Budesonide in 6.8 buffer}}
\label{f-8ef6a5fd052c}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/3e2a0d51-646e-46a3-bbb9-a64deedd8154-upicture10.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/3e2a0d51-646e-46a3-bbb9-a64deedd8154-upicture10.png}{\includegraphics{images/3e2a0d51-646e-46a3-bbb9-a64deedd8154-upicture10.png}}{}
\makeatother 
\caption{\boldmath {Calibration of Budesonide in 7.4 buffer}}
\label{f-809ca120de69}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/de5447a7-96af-4df4-89cd-9890cf86ca51-upicture11.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/de5447a7-96af-4df4-89cd-9890cf86ca51-upicture11.png}{\includegraphics{images/de5447a7-96af-4df4-89cd-9890cf86ca51-upicture11.png}}{}
\makeatother 
\caption{\boldmath {In-vitro release profile of Nanoparticles}}
\label{f-8ac41233e3c1}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/f4103b1a-cd8e-4c71-a14c-0ab0bacbe9a9-upicture12.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/f4103b1a-cd8e-4c71-a14c-0ab0bacbe9a9-upicture12.png}{\includegraphics{images/f4103b1a-cd8e-4c71-a14c-0ab0bacbe9a9-upicture12.png}}{}
\makeatother 
\caption{\boldmath {Particle size Result of Batch F1}}
\label{f-1681446dad14}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/ae2570ac-5b56-4d43-8d70-bfed53802d19-upicture13.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/ae2570ac-5b56-4d43-8d70-bfed53802d19-upicture13.png}{\includegraphics{images/ae2570ac-5b56-4d43-8d70-bfed53802d19-upicture13.png}}{}
\makeatother 
\caption{\boldmath {Particle Size results of Batch F7}}
\label{f-a5d2957af353}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/82c4ea85-27ad-485c-b8b9-299170324f16-upicture14.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/82c4ea85-27ad-485c-b8b9-299170324f16-upicture14.png}{\includegraphics{images/82c4ea85-27ad-485c-b8b9-299170324f16-upicture14.png}}{}
\makeatother 
\caption{\boldmath {Result of zeta potential (Batch F1)}}
\label{f-d1909adb1387}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/01a6d702-9190-479a-aa8d-a8b6fd1284f4-upicture15.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/01a6d702-9190-479a-aa8d-a8b6fd1284f4-upicture15.png}{\includegraphics{images/01a6d702-9190-479a-aa8d-a8b6fd1284f4-upicture15.png}}{}
\makeatother 
\caption{\boldmath {Result of zeta potential (Batch F7)}}
\label{f-8e0e0ad51bd4}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/6a707725-d3bd-4faf-b6c1-2b19d91bc02f-upicture16.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/6a707725-d3bd-4faf-b6c1-2b19d91bc02f-upicture16.png}{\includegraphics{images/6a707725-d3bd-4faf-b6c1-2b19d91bc02f-upicture16.png}}{}
\makeatother 
\caption{\boldmath {Drug release of capsule containing Nanoparticles of batch F1 in triplicates}}
\label{f-a648ad101d71}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/9e803b8d-7c48-43b5-8fc6-ffa9477dd9c7-upicture17.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/9e803b8d-7c48-43b5-8fc6-ffa9477dd9c7-upicture17.png}{\includegraphics{images/9e803b8d-7c48-43b5-8fc6-ffa9477dd9c7-upicture17.png}}{}
\makeatother 
\caption{\boldmath {Scanning Electron Microscopy of Batch F1}}
\label{f-b9823ae4e15c}
\end{figure*}
\egroup
\textbf{Infra-Red spectra}

\textbf{Infra-Red spectrum of Budesonide}

IR spectra of Budesonide were taken on IR spectrophotometer by simply placing small amount of drug in powder form on selenium bromide crystal. In a spectrum, peaks were found at 3497.81 for O-H stretching, 2942.03 for C-H stretching.cm\ensuremath{^{-1}}. IR spectrum of Budesonide was found as given in Figure~\ref{f-ea9afac03b31}.

The Budesonide pure drug shows functional peaks at 1719.22 cm\ensuremath{^{-1}} hence the (-C=O group) confirmed and at 1618.20, 1242.31, 1237.91, 1068.48cm\ensuremath{^{-1}}) shows functional group Ethylene (C=C), carbon-carbon (C-C) Bending, Ether (C-O-C) Bending, Argon (Ar-O-R) Stretching. On the basis of above frequency conclude that Budesonide as a pure drug.

\textbf{FT-IR spectra of Eudragit S100}

FT-IR spectra are used for functional group identification in compound. A small amount of Eudragit S100 in the form of powder was placed on selenium bromide crystal and spectrum was run. The Infra-Red spectra of Eudragit S100 are as depicted in Figure~\ref{f-417f9c099526}.

The IR peaks of Eudragit S100 were compared with standard graph of Eudragit S100 and found to be similar. 

\textbf{FT-IR spectra of Poloxamer}

A small amount of Poloxamer in the form of powder was placed on selenium bromide crystal and spectrum was run. The Infra-Red spectra of Poloxamer are depicted in Figure~\ref{f-cb21793ab7e6}.

Infra-Red peaks for Poloxamer were compared with standard graph of Poloxamer and found to be similar.

\textbf{DSC analysis}

The DSC thermogram of Budesonide was obtained to evaluate the thermal behaviour of pure drug in as given in Figure~\ref{f-8a1fb96ff16d}. Differential Scanning Colorimetric is a thermo-analytical techniques. A calorimeter gives endothermic and exothermic peaks which indicate heat into or out of test sample of DSC thermograms of Budesonide. Changes in the shape of the peak, shift of peak, absence of endothermic and/or exothermic peaks, generation of new peak shows there is interaction.

Differential Scanning Colorimetric (DSC) was carried out on pure Budesonide. DSC is performed and scan was recorded by keeping parameters like heating rate of 10\text{\textordmasculine}C/min and window kept for temperature was 30 \text{\textordmasculine}-300\text{\textordmasculine}C. In this scan reference used was empty standard aluminium pan.

The DSC peak of Budesonide shows sharp endothermic peak onsets from 259.14 and ends at 262.corresponds to standard DSC thermogram of Budesonide.

\textbf{Compatibility by FT-IR Study}

It is necessary to study the compatibility of the excipient with drug. Here drug and excipient is compatible or not was determined with the help of IR spectroscopy. The relative amount of drug and excipient is as given in Table~\ref{tw-39334c4f6b32}.

The compatibility study was performed at the temperature of 55\ensuremath{^\circ}C \ensuremath{\pm}2\ensuremath{^\circ}C and duration for study was 14 days. Individual drug and Drug: Excipient (as ratio 1:1) was kept in vial which was sealed properly. IR scan were taken for drug, polymer and surfactant prior to initiation of study and these raw materials were kept in vials for the ratio and duration as mentioned above. These glass containers were checked for tests like liquefaction, gas formation, caking and colour change if any. IR scan was taken at last after 14 days of study, spectra given in below Figures~\ref{f-341f93cda9ac} and~\ref{f-5f4d211c11e8}.

Both physical and chemical Compatibility study was carried out in the absence of moisture at 40\text{\textordmasculine} C in hot air oven for 14 days and found it compatible. 

\textbf{Infra-Red spectrum of mixture}

Infra-Red spectrum of Drug+ Eudragit and Drug+ Poloxamer are as given in  Figures~\ref{f-341f93cda9ac} and~\ref{f-5f4d211c11e8}  respectively. 

 Drug + Eudragit S100

\textbf{Drug+Poloxamer}

The drug-excipient mixtures were observed for physical incompatibilities such as colour change, liquefaction, caking, and Gas formation and chemical incompatibilities with the help of FT-IR study. The results obtained are as given in Table~\ref{tw-3a4517d34f62}.

\textbf{Formulation }

To prepare polymeric Nanoparticles with a required particle size and to attain a constant formulation, the effect of various process variables was investigated. Variety of organic solvents which could dissolve Eudragit S100 and Drug, i.e. ethanol, methanol, acetone and dimethyl sulfoxide (DMSO) were screened for preparation of Nanoparticles \unskip~\citep{687426:16266206,687426:16266164}. Ethanol was finalized as organic phase based on initial screening and was used to dissolve Eudragit S 100 and Drug \unskip~\citep{687426:16266314,687426:16266077}.\ensuremath{^{\textbf{\space }}}

Nanoparticles were prepared by using modified Nano precipitation method using probe sonication. Required amount of Budesonide and Eudragit S100 were weighed properly and dissolved in Ethanol as organic phase. Then this solution was added drop wise into the water containing Poloxamer as aqueous phase under probe sonication for 10 - 15 min. The nano suspension was then subjected to rotavapour for removal of free ethanol. The above mixture placed in to oven for drying the Nanoparticles.\textbf{\space }

On the basis of literature survey and laboratory work the Nano precipitation method was selected for the preparation of Nanoparticles. Preliminary batches were prepared to whether Nanoparticles formed for this combination or not and if formed at what will be concentration and ration of drug, polymer and surfactant. \ensuremath{^{}}

With the preliminary knowledge we understand that for preparation of nanoparticles it is necessary to select the proper ratio of the Budesonide and Eudragit S100, it was determined by taking ratio such as 1:1 and 1:2 and 1:3 likewise respectively by taking further given in  Table~\ref{tw-8d3a15514c51} and which selected on the basis of extensive literature survey for preparation of colon specific nanoparticles having high entrapment efficiency (EE) and optimized particle size. This method is based on trial and error with the base of preliminary batches results. \ensuremath{^{}}

The amount of Budesonide was kept constant and the ratio of drug to polymer and drug to surfactant was varied as mentioned in Table~\ref{tw-8d3a15514c51}  table. All the batches were manufactured according to experimental design described previously. In the batch F1, F4 and F7 the particles were formed so the batch F1, F4 and F7 was selected for entrapment efficiency and drug release study. 

\textbf{Evaluation of Nanoparticles}

\textbf{Drug entrapment efficiency}

\textbf{\space }Encapsulation efficiency of Nanoparticle was determined by separating free budesonide from budesonide suspension by centrifuging at 5000 rpm for 15 min \unskip~\citep{687426:16265672,687426:16266313}. The clear liquid (supernatant) was taken and diluted with ethanol to calculate un-entrapped drug absorbance measured at 254 nm to calculate encapsulation efficiency. The percentage of drug entrapment efficiency was calculated using subsequent Equation \unskip~\citep{687426:16266164,687426:16265755}

Encapsulation efficiency \%
\begin{eqnarray*}=\frac{Total\;drug\;-free\;drug\times100\;}{Total\;drug} \end{eqnarray*}
\textbf{Drug release study}

The drug release study of budesonide nanoparticles was carried out in dissolution test apparatus II (TDT 08 L Electrolab) at a speed of 50 rpm which contains 900 ml medium at 37\ensuremath{^{0}}C. The capsules are transferred in the medium and samples were taken at certain time interval and analysed by UV at 254nm. The constant dissolution was used by simulating gastric conditions of the gastrointestinal area. For dissolution capsule placed in 700 ml of 0.1 N HCL (pH1.2) for 2 hours. At the end of 2 hours 200 ml of tribasic sodium phosphate was added to dissolution vessel pH adjusted to 6.8 for 3 hrs and finally 7.4 by using 2M NAOH \unskip~\citep{687426:16266314}. 

\textbf{ Particle size determination}

Nanoparticle formulation (1ml) was diluted with 10 ml deionized water in a beaker with constant stirring using a glass rod. The resultant solution was then subjected to particle size analysis. With the help of Dynamic light scattering (DLS) technique, using a zetasizer (Nano ZS, Malvern Instruments, UK) the droplet size was calculated \unskip~\citep{687426:16265672,687426:16266313}.\ensuremath{^{\textbf{\space }}}

\textbf{Zeta potential}

Zeta potential of the formulation was determined with laser diffraction analysis.

\textbf{\space }Resultant solution was then subjected to particle size analysis. Zeta potential measured by Dynamic light scattering (DLS) technique with a zetasizer (Nano ZS, Malvern Instruments, UK) \unskip~\citep{687426:16265672,687426:16266313}.\ensuremath{^{\textbf{\space }}}

\textbf{Scanning Electron Microscopy}

Scanning electron microscopy analysis of the nanoparticle formulation was performed to understand surface morphology of Budesonide loaded nanoparticles.
    
\section{Results and Discussion}
\textbf{Determination of absorption maxima (\ensuremath{\lambda }\ensuremath{_{max}}) and preparation of calibration curves}

The calibration curve for Budesonide using double beam UV spectrometer

The calibration curve for Budesonide in Ethanol, 0.1 N HCl, pH 6.8 and pH 7.4 buffers was prepared by plotting absorbance versus concentration at practically obtained \ensuremath{\lambda }\ensuremath{_{max}} 254 nm. Calibration curve was plotted in replicate manner. Ranges selected for concentration were of 10, 20, 30, 40, 50 \ensuremath{\mu }g/ml. The calibration curve of Budesonide in 0.1 N HCl of pH 1.2 and phosphate buffer of pH 6.8, pH 7.4 were found as shown in Figures~\ref{f-22f11909349d}, \ref{f-c029e03bfb41}, \ref{f-8ef6a5fd052c} and~\ref{f-809ca120de69}  respectively.

From the calibration curve of the Budesonide the regression was found to be 0.947 in 1.2 pH HCl buffer and 0.945 in 6.8 pH phosphate buffer and 0.964 in pH 7.4 buffer. The regression shows that the drug was pure.

\textbf{Percent Entrapment Efficiency (EE)}

The range of percent EE was established from 80\% to 90 \%. EE of optimized formulation was found to be 89.52\%. Trial batches result prove that, polymer concentration shows positive influence of the entrapment efficiency as batch F1, F4, F7 show \% EE 89.52\%,81.30\%,85.50\% have more polymer content.

\textbf{Drug release profile (In-vitro) of optimized batches}

 The drug release profile was also performed on optimized batches (Batch F1, F4 and F7). Release profile was studied for twelve hours. The release curve is shown in Figure~\ref{f-8ac41233e3c1}.

Above data clearly show that the batch F1 and F7 show good value of In-Vitro Drug Release and entrapment efficiency so, two batches F1 \& F7 were taken for further evaluation for determination of particle size and zeta potential.

\textbf{Determination of particle size}

Particle size analysis of nanoparticles for the optimized batches F1, F7 is 318.8 nm and 523.4 nm shown in Figures~\ref{f-1681446dad14} and~\ref{f-a5d2957af353} respectively.

\textbf{Zeta Potential Determination}

Zeta potential of Nanoparticles from optimized batches F1 and F7 is shown in Figures~\ref{f-d1909adb1387} and~\ref{f-8e0e0ad51bd4}  respectively.

\textbf{Filling of Nanoparticles in capsules}

On the basis of experimental design and evaluation of the Nanoparticles optimized batch i.e. F1 batch Nanoparticles were filled in capsule i.e. final dosage form and these filled capsules drug release profile were tested up to 12 hours in triplicate. Drug release data of capsule containing nanoparticles is shown in Figure~\ref{f-a648ad101d71}.

From the dissolution study it was clear that batch F1 shows the good drug release in colonic region.

The batch F1 A, F1 B and F1 C contains Eudragit S100 in 100mg. The result was good because it shows less than 11\% drug release in stomach i.e. 9.01\%, 9.88\% and 9.11\% respectively which is advantageous and achieves our objective to target colon.

\textbf{Scanning Electron Microscopy}

The Scanning Electron Microscopy\textbf{\space }analysis of optimized batch (batch F1) demonstrates that nanoparticles are round in shape. The Budesonide entrapped Nanoparticles do not show presence of cavities. The results of Scanning in Electron Microscopy are shown in Figure~\ref{f-b9823ae4e15c}.

\textbf{Stability study}

The capsule containing Nanoparticles were stored in a container at 25\text{\textordmasculine}C \ensuremath{\pm}5\text{\textordmasculine}C and flooded solution of sodium chloride to attain relative humidity of 60\% \ensuremath{\pm}5\%. Study was continued for a duration of 6 month and time points for testing were 1 month, 3 months and 6 months. This optimized formulation was tested to determine whether there is any significant change in appearance and other evaluation parameters like drug release.

Appearance and in-vitro drug release shows there is no significant change in the results comparable with initial results of optimized batch i.e. Batch F1.
    
\section{Conclusions}
In the treatment of IBD targeted delivery to site of action is advisable, so in this regard considerable progress has been made and novel technologies like nanoparticles were approached. In current study we developed polymeric nanoparticles of Budesonide to target colon in the treatment of colitis. The optimized nanoparticles formulated to minimize drug release in the stomach and release at site of treatment i.e. in colon in treatment of IBD. In vitro release profile demonstrated maximum release of budesonide from the formulated polymeric Nanoparticles. The present study has brought our potential of polymeric Nanoparticles formulation (Nanoparticles in capsule) in ulcerative Colitis. Extensive clinical studies will further substantiate the merit of this novel formulation.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
